Author’s response to reviews

Title: Veno-venous ECMO: a synopsis of nine key potential challenges, considerations, and controversies

Authors:

David B Tulman (dbtulman@gmail.com)
Stanislaw P Stawicki (stanislaw.stawicki@osumc.edu)
Bryan A Whitson (bryan.whitson@osumc.edu)
Saarik C Gupta (saarik.gupta@gmail.com)
Ravi S Tripathi (ravi.tripathi@osumc.edu)
Michael S Firstenberg (msfirst@gmail.com)
Don Hayes (don.hayes@nationwidechildrens.org)
Xuzhong Xu (xuzhong@263.net)
Thomas J Papadimos (thomas.papadimos@osumc.edu)

Version: 5  Date: 20 June 2014

Author’s response to reviews:

Dear Editors,

There were three requests made and they have been completed.
1. Dr. Xu has submitted his agreement with the author order (three times according to him).
2. We have been in a comment about consent and it is right before author contributions.
3. The manuscript conforms with journal format.

Thomas J. Papadimos